Author's response to reviews

Title: Participation of older newly-diagnosed cancer patients in an observational prospective pilot study: an example of recruitment and retention

Authors:

Martine TE Puts (martine.puts@mail.mcgill.ca)
Johanne Monette (johanne.monette@mcgill.ca)
Veronique Girre (veronique.girre@curie.net)
Christina Wolfson (christina.wolfson@mcgill.ca)
Michele Monette (michele.monette@mail.mcgill.ca)
Gerald Batist (gerald.batist@mcgill.ca)
Howard Bergman (howard.bergman@umontreal.ca)

Version: 4 Date: 30 July 2009

Author's response to reviews:

Dear Editor,

I am very sorry that the revisions are late as we had trouble accessing my Universite de Montreal email inbox.

Please find enclosed the revised manuscript with the format changes.

With kind regards,
Howard Bergman